Contents lists available at ScienceDirect

Aquaculture

journal homepage: www.elsevier.com/locate/aguaculture

Lice shielding skirts through the decade: Efficiency, environmental interactions, and rearing challenges

through the production cycle.

Kristbjörg Edda Jónsdóttir^{*}, Andreas Ugelvik Misund, Leif Magne Sunde, Merete Bjørgan Schrøder, Zsolt Volent

SINTEF Ocean, NO-7465, Trondheim, Norway

ARTICLE INFO	A B S T R A C T
<i>Keywords:</i> Lice shielding skirts Barrier technology Fish welfare Louse management Cage environment	Lice shielding skirts are a preventative measure against salmon lice in Atlantic salmon farming. The skirt is wrapped around the top meters of the net cage to divert the current flow around the cage, and thereby keep the salmon lice out. Despite these skirts being used actively in Norwegian aquaculture for the past decade, there is no standardised way of using them, and therefore type, depth and operating procedures vary between sites. The academic literature on the lice shielding efficiency of these skirts is not extensive and reported efficiency varies across studies and sites with some reporting favourable results, while others find none. Some also report of

The aim of this paper is to provide a comprehensive overview and summary of relevant academic and grey literature from the last decade to identify knowledge gaps that must be filled to achieve optimal use of lice shielding skirts. This paper focuses on three main topics: lice shielding efficiency, interaction with the current flow and rearing challenges. The positive results from some sites indicate that skirts have potential as a tool against salmon lice, however, to create a best practice recommendation for skirt use, more knowledge is necessary on the interaction between skirt and the environment, and sufficient monitoring procedures and decision-making tools must be established.

welfare related issues, with dissolved oxygen levels being the most prevalent, but this too varies across sites and

1. Introduction

The parasitic salmon lice (Lepeophtheirus salmonis) are a naturally occurring ectoparasitic crustacean in the North Atlantic and North Pacific Oceans (Finstad et al., 2000; Thorstad et al., 2015; Misund, 2019). These lice are a potential threat for wild salmon, a welfare issue for farmed Atlantic salmon (Salmo salar) (Finstad et al., 2011; Thorstad et al., 2015; Forseth et al., 2017), and a significant financial burden for the aquaculture industry, costing the Norwegian aquaculture industry alone approximately 5 billion NOK in 2015 (Abolofia et al., 2017; Iversen et al., 2019; Brooker et al., 2018).

The welfare issues and the high cost in aquaculture have resulted in several different approaches for controlling the lice, which can roughly be divided into three groups: immediate, continuous, and preventative measures (Svåsand et al., 2017; Coates et al., 2021). For the past four decades, immediate chemotherapeutants have been the dominant method (Overton et al., 2019). However, the use is decreasing due to an increase in chemotherapeutants-resistant lice (Aaen et al., 2015) and unwanted environmental impacts such as harming nearby shrimp populations (Bechmann et al., 2019, 2020). As a result, there has been an increase in use of non-chemical immediate treatments such as thermal, fresh water or mechanical (Overton et al., 2019).

Thermal, fresh water and mechanical treatments require handling of the fish in the form of crowding and pumping, which are sources for injuries to the gills, skin, fins, and snout (Svåsand et al., 2017; Noble et al., 2018). Increased handling is also linked with an increased risk of salmon escapes (Thorvaldsen et al., 2015) and increased mortality rates for specific treatments (Overton et al., 2019). For instance, the high mortality rate of 15.5% in the Norwegian salmon industry in 2021 is partially attributed to an increase in the number of mechanical and thermal treatments (Sommerset et al., 2022). An alternative to the immediate treatments is continuous non-chemical treatments that do not require any handling, such as cleaner fish (Imsland et al., 2014, 2018). However, there are strong concerns regarding the welfare of cleaner fish

* Corresponding author. E-mail address: kristbjorg.jonsdottir@sintef.no (K.E. Jónsdóttir).

https://doi.org/10.1016/j.aquaculture.2022.738817

Received 29 June 2022; Received in revised form 29 August 2022; Accepted 8 September 2022 Available online 12 September 2022

0044-8486/© 2022 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

Review







(Geitung et al., 2020; Overton et al., 2020; Garcia de Leaniz et al., 2022) and others have also questioned their efficiency (Overton et al., 2020; Barrett et al., 2020a).

Given the negative aspects of the immediate and continuous measures, Barrett et al. (2020b) made an argument for a more preventionfocused louse management. Preventive measures aim to reduce the contact between lice and fish and are the preferable approach economically, environmentally and from a fish welfare perspective (Barrett et al., 2020b). The wide variety of preventive measures can be characterized as heterogenous, and include measures such as on-land production, closed cages, selective breeding, fallowing, functional feeding, geographic spatiotemporal management, manipulation of swimming depth, repellents, and host cue masking (Barrett et al., 2020b).

Included in the preventative measures are the barrier solutions, such as the lice shielding skirts and the "snorkel" cages (Dempster et al., 2009). Shielding skirts are a barrier measure that tries to keep the lice out of the cages by blocking their access to the upper layers of the water column in the cage (Fig. 1). Snorkel cages are also a barrier measure as it keeps the salmon at deeper depths by using cages with net roofs limiting the salmon's access to the upper water column. To ensure that the salmon can refill their swim bladder, the snorkel cages have a vertical chamber at the centre of the cage, known as the snorkel, providing access to the surface. To protect the salmon from any lice when traversing this area, the snorkel is typically tarpaulin-enclosed, hence combining the two methods (Stien et al., 2016; Oppedal et al., 2017; Wright et al., 2017; Geitung et al., 2019). The use of "snorkel" cages is still quite limited with only a handful of commercially scaled cage trials (Geitung et al., 2019). Shielding skirts on the contrary are today one of the most established "barrier" technologies in Norway and is also used internationally.

Initially, impermeable tarpaulin skirts were mounted around the cage's sides to increase efficacy of chemotherapeutants during bath treatments against salmon lice (Whyte et al., 2016; Overton et al., 2019). These semi-enclosures did not cover the bottom of the cage net, hence the exact mixture concentration and diffusion of chemotherapeutants were unknown and random due to turbulence and inflow of water, further exacerbating the issues with effectiveness and chemotherapeutant-resistant salmon lice (Volent et al., 2017). In 2011, a new regulation in Norwegian aquaculture stipulated that it should be compulsory for fish farmers to ensure full enclosure of the net-cage during in-cage delousing treatments (Norwegian Ministry of Trade, Industry and Fisheries, 2012). Almost concurrently with these stricter requirements for bath treatments, lice skirts were trialled as a preventive measure (Coates et al., 2021). According to Næs et al. (2012), the concept of using permeable skirts as a preventive measure to avoid salmon lice infestation were discussed back in 2005, approximately 6-7 years before the first trials began in 2011-12.

Shielding skirts today come in several different fabrics and design. The two main categories are impermeable and permeable skirts. Impermeable skirts are typically made of a tarpaulin cloth that block all incoming currents. Permeable skirts are made of a fine mesh fabric with a solidity of roughly 50% that in theory will allow some of the water to pass through while keeping the lice out. Most lice skirts when installed consist of either one large piece with overlap, or two pieces with an overlap on each side of the cage. To assure that the skirts maintain a vertical position in the water column, sinkers are attached to the lower parts of the skirts.

In addition to the skirts made of fabric, some newer cage designs have opted to directly implement "barrier" technology in their designs. For instance, Aquatraz (Midt-Norsk Havbruk AS) has a skirt installed along the upper 18 m of a steel cage structure (Midt-Norsk Havbruk, 2020), and Havfarm 1 (Nordlaks AS) was originally designed to have a steel skirt installed (Jenssen, 2016). There is understandably a limited amount of literature on these new production systems since they are currently under trial, hence they will not be discussed further in this paper.

How skirts are used varies widely between sites. In a small-scale qualitative study, respondents from the Norwegian aquaculture industry shared their anecdotal knowledge and experiences with lice skirts (Misund et al., 2020). The responders indicated that the use of skirts is highly dependent on the environmental conditions at the production site, and that there are different practices between sites regarding choice of skirt depth, when and how to mount, and in which period to use the skirts (Misund et al., 2020). There is as of today no best recommendation available for the use of skirts.

In the available literature surrounding barrier technology there are hardly any sources documenting the use of skirts in other salmon producing countries than Norway and Scotland, and there are no scientific publications on the subject from Scotland yet (A. Currie, personal communication, September 1, 2021). In general, the academic literature is not extensive, and a large portion of the results from full-scale trials are presented in project reports and other grey literature. The aim of this paper is to give a comprehensive overview of the literature from the last decade, both academic and grey literature. This paper is structured into six parts. Following the introductory section, Section 2 describes the design basis of lice shielding skirts on the premise of salmon lice's biology and behavioural traits. Section 3 discusses the documented lice shielding efficiency of the skirts. Section 4 describes the interaction between skirt and current flow, including current flow reduction, changes to current flow pattern, impact of hydrographical conditions and structural implications. Section 5 describes how the interaction between the skirt and the environment influences the internal rearing environment of the cage. Finally, Section 6 emphasizes the knowledge gaps and what future work is necessary to achieve optimal use of lice shielding skirts.



Fig. 1. Illustration of lice-shielding skirts mounted on a cylindrical (left) and conical (right) cage. The skirt on the left is installed as one long piece of fabric, with an overlap in the front.

2. Salmon lice behavioural traits and biology as a premise for barrier technology

The core concept of the lice skirt is to utilise the behavioural traits of the salmon lice to keep them out of the cage volume, specifically their vertical position in the water column. However, the lice's vertical position depends on several variables, including its development stage.

The salmon lice develop through eight life stages, starting with the planktonic non-parasitic nauplius 1 & 2 larvae, before developing into the planktonic infective copepodid stage (Hamre et al., 2013; Crosbie et al., 2020). It is during the infective stage that the lice attach itself to its host and begins to feed of the skin, mucus, and blood (Costello, 2006). The dispersion of the nauplii and copepodid stages is mainly dictated by the ocean currents as it occurs through passive transport (Costello, 2006). Their vertical position on the contrary is controlled through locomotion and sinking (Allen and Lewis, 2013; Crosbie et al., 2019), and is strongly influenced by their environmental preferences.

The vertical migration pattern of nauplii and copepodids in experimental columns has several distinctions. When a salinity step, a halocline, is present both nauplii and copepodid stages aggregate near the halocline depth (Heuch et al., 1995; Crosbie et al., 2019), but they react differently to salinity strength. The non-parasitic nauplii stages avoid salinities below 30 ppt, while the parasitic copepodids are found in water with salinities as low as 16 ppt (Heuch et al., 1995; Crosbie et al., 2019). The nauplii also prefer colder water temperatures, while the copepodids show no preference for temperature (Crosbie et al., 2020).

The two stages show different reactions to light. Early results indicated that copepodids were generally found deeper during the night than the day, while nauplii were less active and less responsive to light (Heuch et al., 1995). However, more recent studies indicate that nauplii also respond to light. When nauplii and copepodids were exposed to light ranging from 0 to 80 μ mol m⁻² s⁻¹ for up to 12 h at a time, the nauplii had a stronger response to increasing light intensities, aggregating at the surface (Szetey et al., 2021). The copepodids on the other hand swam upwards independently of light intensity and duration (Szetey et al., 2021).

There is also evidence of differences between lice families. Copepodids are reactive to changes in pressure, swimming upwards towards the surface when the pressure is increased in test tubes (Coates et al., 2020). However, the response of copepodids varied between families, and results suggest that depth response of copepodids may have a genetically inherited basis (Coates et al., 2020).

The different behavioural traits of the nauplii and copepodid indicate that copepodids are motivated by host-finding, while nauplii are driven by survival and dispersal (Szetey et al., 2021). Presumably due to the stage's behavioural drivers and environmental preferences a higher abundance is found in the surface layers during the day (Heuch et al., 1995; Hevrøy et al., 2003). The concept behind "depth-based" and "barrier" technologies has therefore been to minimise host-parasite interaction, specifically by exploiting the vertical migration pattern of the copepodids and their preference for the upper water layers.

3. Lice shielding skirt efficiency

Despite widespread use of lice shielding skirts, scientific documentation of their effects on lice infestations and fish welfare are lacking. There are a few academic publications, but a large portion of available material is in the form of grey literature: project reports, reports from the manufacturers or brief reviews in annual reports from academic institutions (Stien et al., 2018).

When reviewing the literature, the results are not consistent, with some claiming near to no effect of the lice shielding skirts, while others document great results (Table 1). In the first report on the efficacy of permeable lice shielding skirts by Næs et al. (2012), 3 of the 6 cages included in the experiment were equipped with 10 m deep skirts from May until November 2011. The results were promising, with 70% less lice per fish in the shielded cages than in the unshielded (Table 1), however the infestation pressure was relatively low during this study with an average of 0.05 sessile lice during the entire period in the unshielded cages (Næs et al., 2012).

In the following report by Næs et al. (2014) permeable skirts of 6 and 10 m were installed at 6 sites. The duration of the study varied from 2 to 7 months. Five of the sites had both cages with skirts and control cages without skirt, and the last site had skirts installed on all the cages. For the farms where lice were detected, there was a decrease in both sessile and mobile lice when using skirts. Grøntvedt et al. (2018) utilised part of the dataset from Næs et al. (2014), and estimated the weekly reduction of pre-adult and adult male lice to 30% when using permeable plankton

Table 1

Overview over studies (both peer-reviewed and grey papers) that have studied the impact of skirts on lice infestation levels. All studies have used different approaches in finding lice shielding efficiency, with some comparing lice per fish in cages with and without, and others using regression analysis to determine the impact of the skirt. It is the general trend which is of most importance, but max-min reduction is included to give an indication of range of results. Studies with several sites may have different results at each site, so if there are cases of both a reduction and no effect, this is included in the table. The rows containing data from peer-reviewed studies are written in bold and are italicized. The other studies are grey-literature.

			Skirt and preventative measures			Trend lice shielding		
Study	# sites	Duration	Туре	Depth [m]	Additional measures	Reduction/No effect	Comparison method	Max./Min Reduction [%]
Næs et al. (2012)	1	30 wks (2011)	Permeable	10		Reduction	With/Without same site	- / 70
Næs et al. (2014)	6	10–14 wks (2012- 2013)	Permeable	6 & 10		Reduction	With/Without same site	7 / 83
Grøntvedt and Kristoffersen (2015)	17	9–33 wks (2013–2014)	Permeable	5	Cleaner fish	Reduction/ No effect Reduction/ No	With/Without same site Between sites	6 / 28 0 / 80
Stien et al. (2018)	1	15 wks (2014)	Permeable	10		Reduction	With/Without same site	- / 82
Grøntvedt et al. (2018)	5	11–21 wks (2012-2014)	Permeable	6 & 10		Reduction/ No effect	With/Without same site	30% weekly
Midtlyng et al. (2019)	3	28–66 wks (2017–2019)	Impermeable*	5,8&9	Cleaner fish,	No effect**	With/Without same site	0 / 50**
Bui et al. (2020)	1	56 wks (2016–2017)	Permeable	6	Cleaner fish, functional and submerged feeding, submerged lights	Reduction	With/Without same site	- / ~40

* P.J. Midtlyng, personal communication, December 8, 2021.

* With the exception of a short spike during the spring. See text for more info.

skirt.

Other surveys on the use of permeable skirts reported an average decrease of 18% lice infestation for cages with skirts compared to cages without on the same farm (Grøntvedt and Kristoffersen, 2015). And when inspecting the difference between neighbouring farms, where some farms used skirts and others did not, the average reduction in lice infestation was 54% (Grøntvedt and Kristoffersen, 2015). It should however be noted that there were 17 locations included in the study, and the variation of average reduction in lice infestation compared with neighbouring farms varied between 0% to 80%, meaning some farms had no effect of using lice shielding skirts compared with their neighbours (Table 1).

In the scientific study by Stien et al. (2018) a reduction of 82% during a period of 3 months was reported. From May to September in 2014 three of the farm's 10 cages had 10 m deep permeable skirts installed, like those used in Næs et al. (2014). The largest effect of the skirt was seen during the last month of deployment with an average of 0.18 (+/- 0.06) lice per fish in the cages with skirt, and 1.03 (\pm 0.19) lice per fish in the cages without skirt.

However, Midtlyng et al. (2019) saw no effect when using skirts, as did one of the locations in Grøntvedt et al. (2018). In Midtlyng et al. (2019) four locations were mounted with skirts of different lengths for up to a year. On one of the four locations with a 5 m deep skirt a sudden peak in lice infestation indicated that the cages with skirts had a reduction of 50% of lice compared to the cages without skirt. However, the authors concluded that it was not possible to ensure that this result was due to the skirt and not the location of the cages (Midtlyng et al., 2019). As the farm was exposed to strong currents, only the four cages closest to the feeding barge had skirts installed, so the cages with and without skirts were not randomly assigned. On the other locations in the same report, there was little lice pressure throughout the year, and little difference between cages with and without skirts.

In a 13-month long study by Bui et al. (2020) the effect of combining different passive lice management strategies including lice shielding skirts were studied. The site studied had 12 cages, all with cleaner fish, 9 had in addition functional feeding, 6 had additional deep lights and feeding, and 3 of them had lice shielding skirts. The three cages which had cleaner fish, functional feeding, deep lights, and lice shielding skirts had the lowest rate of new infestations (Bui et al., 2020). Interestingly, the cleaner fish in the skirted cages were the least efficient with the smallest number of lice found in their guts (Gentry et al., 2020). It was hypothesized that this was due to reduced interaction between cleaner fish and salmon as the salmon potentially swam deeper due to the submerged lights (Gentry et al., 2020).

Apart from Midtlyng et al. (2019), all the technical reports and studies inspecting lice shielding efficiency, have used permeable skirts (Table 1). There are no studies comparing impermeable and permeable skirts at the same location, so the results with permeable skirts are not necessarily representative for cages with impermeable skirts. The difference in skirt length at the different sites also complicates comparison between studies.

It has been theorized that deeper skirts would have improved efficiency, as studies show that salmon held at 4–8 m and 8–12 m have a lower infestation rate than salmon held at 0–4 m (Hevrøy et al., 2003), and that deeper "snorkel" cages have less lice abundance (Oppedal et al., 2017). It should however be noted that in Oppedal et al. (2017), there were still lice present in "snorkel" cages of 16 m depth. In Næs et al. (2014) the results from several sites were combined and indicated an average reduction of sessile lice of 49% when using 10 m deep skirts, and 28% when using 6 m skirts. However, in Grøntvedt et al. (2018) which used part of the data set from Næs et al. (2014), it was underlined that there was only one farm which had both 6 and 10 m skirts, hence there was not sufficient data to reveal any significant shielding difference between 6 and 10 m skirts.

Although some of the studies have favourable results, it is difficult to ensure that the reduction in lice at each site was only due to the lice shielding skirt. Comparing the studies is also not possible as some studies utilised cleaner fish and deployed the skirt during different periods and durations. Not all sites began production with skirts installed, and it is not certain that the skirt remained on during the entire time in the different studies. When comparing the maximum and minimum reductions in the different documentation, it is also clear that there is a large discrepancy within each study (Table 1). This could be due to low lice pressure, as this makes it difficult to document statistically significant difference between cages with and without skirts, as highlighted by Midtlyng et al. (2019), or site variations.

The study by Bui et al. (2020) exemplifies the difficulties of studying reduction in lice infestation over a longer period, and how the environment may influence the lice shielding effect. The site of the study was near Vindsvik in western Norway, and through the 13 months study, there were periods with influx of freshwater and salinities as low as 5 ppt. This influx of freshwater may both influence the lice behaviour and preferred depth in the water column (Crosbie et al., 2019), but it could also have altered the current flow pattern at the site reducing the skirts' ability to divert the water.

4. Lice shielding skirt and current flow interaction

4.1. Current flow reduction

When the current moves through a net structure, the current speed is reduced (Løland, 1993; Patursson, 2008; Klebert et al., 2013). This reduction is influenced by the solidity of the net (Bi et al., 2013) and is found through both singular cages (Klebert et al., 2015) and entire farms (Winthereig-Rasmussen et al., 2016). A reduction in current speed has a direct impact on both the environment inside and downstream of the cage. Downstream of the cage, the reduced current speed can influence how particles and micro-organisms such as pathogens and zooplankton are dispersed (Klebert and Su, 2020). Inside the cage, the reduction in current speed can impact the dissolved oxygen levels, as the main source for fresh oxygenated water in an Atlantic salmon farm is the physical transport of water through the cage (Wildish et al., 1993; Johansson et al., 2006). A reduction in current speed through the net will force a portion of the water around the cage, rather than through, meaning that less fresh oxygenated water passes through the cage.

In full-scale trials there is a near linear reduction in current speed when measuring the current upstream, inside, and downstream of an empty cage, with 21.5% reduction inside the cage and 35% reduction downstream of the cage (Klebert et al., 2015; Table 2). This reduction is higher when fish is present, for instance in a stocked small-scale cage with a diameter of 4.6 m, the current reduction inside the cage was as high as 31% (DeCew et al., 2013). Which compares reasonably well with the results from Johansson et al. (2014) where the reduction varied between 0 and 50% when the incoming current speed was 20 cm/s (Johansson et al., 2014).

Unsurprisingly the current speed both inside and downstream of a shielded cage is further reduced, and the reduction is also influenced by present biomass. Downstream of an empty cylindrical cage with 10 m deep impermeable skirts the current speed is reduced by 61% (Klebert and Su, 2020). While downstream of a stocked conical cage with a 10 m deep permeable skirt, the current is reduced by 70.4% (Jónsdóttir et al., 2021b).

In a stocked 50 m diameter cage with and without an impermeable lice shielding skirt, the maximum current speed reduction inside the cage was 91% when the skirt was deployed, and 53% when the skirt was removed (Frank et al., 2015). The current was generally quite slow during the trial, but the largest reduction of 91% was seen when the current speed outside of the cage was 13 cm/s (Frank et al., 2015). These results compare well with a more recent study, where the same cage was studied with and without an impermeable skirt and had a maximum reduction of 78.3% when the skirt was deployed, and 44.3% when the skirt was removed (Jónsdóttir et al., 2021a).

Table 2

Overview of reduction in current speed downstream and inside cages from peer-reviewed studies. Studies written in bold and italics are performed in empty cages with no fish present. Inside cage reductions than span both Min. and Max. reduction shows the average reduction as presented in the studies.

				Downstream	of cage	Inside cage	
Study	Cage diameter	Skirt type	Length	Distance	Max Reduction	Min. Reduction	Max. reduction
DeCew et al. (2013)	4.6 m	-	_	_	-	31%	
Johansson et al. (2014)	41 m	-	-	-	-	0	50%
Klebert et al. (2015)	41 m	-	-	15 m	35%	21.5%	
Frank et al. (2015)	50 m	Impermeable	5 m	-	-	32%	91%
Klebert and Su (2020)	50 m	Impermeable	10 m	60 m	61%	-	-
Jónsdóttir et al. (2021a)	50 m	Impermeable	6.7 m	-	-	35.5%	78%
		-	-	-	-	19.7%	44.3%
Jónsdóttir et al. (2021b)	50 m	Permeable	10 m	25 m	70.4%	60.8%	92.4%

Interestingly, the study with a permeable skirt installed on a conical cage had a higher maximum reduction of 92.4% inside the cage than downstream (Jónsdóttir et al., 2021b; Table 2). Possible explanations for this were the relatively large fish, and high density in the cage, or the conical shape of the cage. The skirt had been installed around the cage like a cylinder, so when the current increased the skirt was pushed into the cylindrical cage. This change in inclination may have reduced the skirts permeable properties, as an increase in inclination of net panels is known to reduce the flow (Bi et al., 2013).

It should be noted that the methodologies used in these studies vary greatly. Where in the cage the current was measured, for how long, size of cage, size of fish, fish density and method used to determine the reduction in current speed differ. There were also large variations in incoming current speed. In Klebert et al. (2015) the minimum current speed was 0.15 cm/s and the maximum exceeded 0.50 cm/s, while in Jónsdóttir et al. (2021b) and Frank et al. (2015) the current did not exceed 0.15 cm/s. However, all these papers conclude that there is an increased reduction in current speed downstream and insides cages with permeable and impermeable shielding skirts compared to cages without. This reduction in current speed could influence the current flow pattern around the skirt, and thereby alter its preventative efficiency.

4.2. Current flow pattern

In addition to influencing the reduction in current speed, the skirt also influences the current flow pattern within the cage. In one of the first reports on the interaction between lice shielding skirts and the current, a simulation of the current flow around a lice shielding skirt was presented (Lien and Høy, 2011). The simulation was of an impermeable skirt with a diameter of 50 m, without fish and net, and with a stable incoming current of 10 cm/s. The skirt was modelled as a rigid skirt, meaning that there was no deformation. The simulation showed that a large portion of the current was forced around the cage, which would have kept the lice out. However, a proportion of the water was also pushed down, beneath the skirt, and up into the back half of the cage, meeting the downstream area of the skirt. The current was there pushed upwards by the skirt and towards the centre of the cage as illustrated in Fig. 2 (Lien and Høy, 2011). This differed to an unshielded cage, where the current would flow straight through the cage without changing direction.

This change in flow direction within the skirt is observed in full-scale studies, with the current direction being relatively stable when no skirt was deployed but varying up to 100° when the skirt was installed (Frank et al., 2015; Jónsdóttir et al., 2021b). There was little deflection of current direction in Jónsdóttir et al. (2021a), where a shorter skirt of 6.7 m was used compared with the 10 m deep skirt in Frank et al. (2015) and Jónsdóttir et al. (2021b). In Stien et al. (2012) the shielded cage had a 3 m deep skirt and the current direction did not deflect through the cage, except for the current speed a couple of meters from the downstream area of the skirt. In this position the current pointed in the opposite direction, agreeing with the simulations presented in Lien and Høy (2011).



Fig. 2. Current flow pattern interaction with skirt. A) shows the horizontal view of the current flow inside the shielded area. B) shows the cross-section of the cage showing how the current flow differs within the shielded and unshielded part of the cage (Illustration: Zsolt Volent, SINTEF Ocean).

Empty full-scale cages with skirts also show indications of the simulated pattern. In two of four trials when measuring the current speed inside an empty shielded cage, the current speed at two different locations within the skirt volume flowed in the opposite direction to the current beneath the skirt depth (Klebert and Su, 2020). Assuming the current speed and direction measured beneath the skirt volume are representative for the main current direction, this indicates that the current behaved as stipulated by the simulation by Lien and Høy (2011).

To investigate if the lice shielding skirts redirected the current around the cage, full scale trials using fluorescent dye were carried out (Frank et al., 2015). Fluorescent dye was released outside a cage without skirt, outside a cage with skirt, and just inside the skirt and then monitored using an aerial camera. The skirt was 5 m deep, and the dye was released at 2.5 m depth. No blocking effect was observed when the skirt was not installed, and the dye passed through the net and into the stocked cage. After the dye had passed through the net wall it took roughly 3 min for the dye to reach the centre of the cage and was still visible inside the cage after 30 min (Frank et al., 2015).

It was theorized that the reason the dye did not simply pass through the cage was the swimming activity of the fish. Specifically, that the torus swimming pattern created a pumping effect, pushing water out at depths of high biomass, and drawing in water from above the below this point (Gansel et al., 2014; Frank et al., 2015). No indication of this pumping effect was seen in the velocity measurements done in Klebert and Su (2020) when no skirt was deployed. However, there was a weak vertical velocity component towards the surface in both stocked and empty shielded cage trials (Klebert and Su, 2020; Jónsdóttir et al., 2021a). As this effect was seen both with and without fish, it was concluded that this was an effect of the skirt, rather than the fish. It is unclear why the dye remained inside the cage in Frank et al. (2015) when the cage was unshielded.

Dye was also released in two trials outside of the cage with the skirt deployed (Frank et al., 2015). The first trial had relatively high velocity, the two-point measurements of speed outside the cage recorded 13.1 cm/s and 9.5 cm/s respectively, while in the second trial the current speed outside the cage was 4.3 cm/s and 7.6 cm/s. In the first trial there was only a small portion of dye that entered the cage, most of it being redirected around the skirt. In the second trial however, a large portion of the dye entered the cage in the upstream part of the cage before traversing the cage (Frank et al., 2015). Despite using the same cage, the results demonstrate how the current flow pattern around a shielded cage can vary from the simulated flow pattern. It is unclear why the results from the two trials differed, however one potential factor that might explain the variation are hydrographical conditions at the site.

4.3. Salinity and density gradients affecting skirt deformation

A temperature or salinity gradient can cause a density difference in the water column. Water with higher salinity and lower temperatures is denser and will position itself beneath water with lower salinity and higher temperatures, causing a gradient in the water column. This is typically caused by either an influx of freshwater from precipitation, nearby rivers, or ice melting, or from the surface water being heated by the sun. A pycnocline is the layer where the density of the water column changes abruptly. Once these layers are established, water moving vertically must overcome an additional buoyancy force added by the density stratification (Imberger, 2012). This can prevent vertical mixing between the layers. It will take more energy to push denser water up into the lighter water layer, hence the presence of a pycnocline will alter how the current flows around the lice shielding skirt (Fig. 3). Furthermore, the layers may behave differently from each other, interacting differently with the skirt depth.

By altering the current flow pattern, a present pycnocline could also influence the lice shielding efficiency, as mentioned in Bui et al. (2020) which observed an influx of freshwater during their study. The presence of a pycnocline is also reported to influence the internal environment conditions of a cage both for unshielded cages (Johansson et al., 2007) and shielded cages (Jónsdóttir et al., 2020), which will be discussed in more detail in Section 5.2. With regards to the study by Frank et al. (2015), it is possible that there was a gradient higher up than the skirt depth. If the dye was then released in the upper layer of the water, it would be difficult to force this lighter water into the heavier water below, forcing more of the water around the cage. However, if the water column was homogenous, more of the water may have been pressed down when reaching the skirt, forcing the dye underneath and into the skirt. Unfortunately, no measurements were done of the temperature and salinity with depth, so it is not possible to determine the cause of the difference.

There are observations of different density within and outside the skirt (Jónsdóttir et al., 2020), and there are reports of sites experiencing deformations caused by these differences. Fish farmers report of lice shielding skirts "Barrelling" or "Hour-Glassing" at farms (Fig. 4). "Barrelling" is when the skirt is forced out in all directions at a certain depth,



Fig. 3. Principles of how a present pycnocline can alter the current flow interaction with a lice shielding skirt. A homogenous water column will allow the water to be pressed down along and underneath the skirt, while the pycnocline may limit the waters motion by preventing the less dense water from moving down into the denser water layer.



 ρ_i = Density inside (kg m⁻³)

Fig. 4. Example of hour-glassing (left) and barrelling (right).

causing the skirt to take on the shape of a barrel, while "Hour-Glassing" is the opposite with the skirt being sucked in all around the skirt at a certain depth causing the skirt to take on an hourglass like shape (Misund et al., 2020). It can be theorized that barrelling is caused by higher density water inside the skirt than outside, while hour-glassing is caused by lighter water inside the skirt than outside, but this has yet to be confirmed, as there are no scientific publications on this phenomenon. Another theory has been the influence of the fish swimming in a circular pattern causing a pumping effect as previously discussed (Gansel et al., 2014; Frank et al., 2015). However, as Klebert and Su (2020) concluded with that there was no indication of a pumping effect, it is more likely that these two shapes are caused by density differences. How these deformations influence the skirt's shielding ability is also unknown.

4.4. Structural implications of water currents: Net deformation and mooring loads

In addition to the current potentially leading more lice into the cage during specific hydrographical conditions, ocean currents can also reduce the lice shielding efficiency by deforming the skirt. Strong currents are known to cause deformations of both cage and skirt, and increase drag forces and mooring loads, making some farms incompatible with skirt use (Lien and Volent, 2012; Lien et al., 2014; Midtlyng et al., 2019).

In the beginning of the skirt era, it was theorized that strong currents could deform the skirt and push the skirt upwards, reducing the blocking efficacy of the skirt, but early studies revealed that this deformation was most likely not an issue. When exposing model cages with tarpaulin skirts equivalent to 4.2, 6.2, and 9.2 m deep in full-scale, with a full-scale current speed of 21 cm/ s, the skirts lifted <10% towards the surface around the entire perimeter (Lien et al., 2014). Another study using the same test facilities as Lien et al. (2014), but skirts equivalent to 5 and 10 m in full-scale, observed similar lifting for 10 m skirt as the 9.2 in Lien et al. (2014) (Volent and Bekkevold, 2017). The shorter skirts in Lien et al. (2014) however were lifted higher than the 5 m skirt in Volent and Bekkevold (2017).

There was also differences in the drag forces measured in the two studies. Drag forces on cages increase with stronger current speed and the presence of a skirt. In Lien et al. (2014) the drag force on cages with skirts was 40% higher than on cages without skirts, while it was only 25% higher in Volent and Bekkevold (2017). The discrepancies between the two studies was most likely caused by different mounting of the skirt. In Lien et al. (2014) the skirts were mounted on the outside of the outer collar. In addition, the deformation of the skirt was obstructed by the chains connecting the sinker tube to the floating collar. In Volent and Bekkevold (2017) the skirts were mounted on the inside of the inner collar closer to the net which gave a smaller diameter of the skirt. It is therefore probable that the results from Volent and Bekkevold (2017) are more representative as today's practice is to mount the skirts on the inside of the floating collar.

Additional tests were carried out in Volent and Bekkevold (2017) to document how different parameters affect skirt deformation. The results showed little differences in deformation between permeable and impermeable skirts mounted on the cylindrical net. However, skirts mounted on a conical shaped net had larger deformation at low current speeds compared with the skirt mounted on a cylindrical net. This was because the skirt could freely move from its vertical position until the skirt touched the net wall (Volent and Bekkevold, 2017).

Adding weight to the bottom of the skirt had also little effect on the deformation of the skirt. Experiments with 100 kg weight mounted in one point upstream of a 10 m impermeable skirt had no effect on vertical displacement of the skirt for current up to 21 cm/s. However, for current speeds of 41 cm/ s the weight had an opposite effect than intended on the deformation; the skirt depth in front was lifted 38% higher with the heavy weight compared to tarpaulin without weight. Similar deformation results were observed with a heavy weight downstream. The results lead to a recommendation to use 5–8 kg m⁻¹ weight at perimeter at the bottom of the skirt to prevent the skirt from floating up to the surface caused by air bubbles or waves (Volent and Bekkevold, 2017).

Full-scale experiments agree well with these results. In Volent and Jónsdóttir (2019), the deformation of the skirt at three full-scale cages at different sites were compared with the lab results in Volent and Bekkevold (2017). The three sites Fornes, Josommarset and Korsneset were all equipped with pressure tags at the bottom of the skirt positioned upstream and downstream of the main current direction (Table 3). The comparison between model experiment and full-scale measurements at Fornes showed good correlation for the tags upstream of the cage, but not downstream. A possible explanation for this was the location of the skirts overlap.

Lice shielding skirts consists of either 1 or 2 parts, where the ends must either overlap with 5–10 m when mounted or be stitched together. The overlap was not stitched together and was placed downstream, which allowed more deformation than in the model experiment where the overlap was place 90° to the current direction. At Josonmarset the correlation was good at the downstream side, but the skirt was lifted a little higher in front compared with the lab studies. At Korsneset both the front and rear part of the skirt was lifted higher than expected (Volent and Jónsdóttir, 2019).

Unlike the deformation by currents, waves in model-scale tests caused the skirt, independently of length and type, to be lifted equivalently to the wave height (Volent and Bekkevold, 2017). There are currently no full-scale studies on the deformation of skirts in waves. More studies are needed here to inspect the interaction between waves and skirts. The lifting of skirts caused by currents and waves upstream could cause more water to pass underneath the skirt and into the cage, reducing the shielding efficiency.

5. Shielding skirts and rearing challenges: Implications for fish welfare and production

5.1. Challenges of the shielded internal cage environment

Lower dissolved oxygen, poor water quality (plankton), increased

Table 3

Table shows information for each site studied in Volent and Jónsdóttir (2019), which are were compared with the results of Volent and Bekkevold (2017).

	Nordlaks – Fornes	Ellingsen – Josommarset	SalMar/ACE – Korsneset
Cage and fish	Cage perimeter 157 m Plankton skirt 10 m deep	Cage perimeter 100 m	Cage perimeter 157 m Impermeable skirt 7 m
	Conical net 55 m deep	Impermeable skirt 5 m	
	Fish size 0,1 kg	Fish size 1,6 kg	Fish size and biomass not available
	Biomass 190 metric ton	Biomass 238 metric ton	

particle accumulation and deterioration of gills, poor or no effect on salmon lice infestation, perceived increase in amoebic gill disease (AGD) infestation or perceived lower feed conversion ratio (FCR) are amongst the primary reasons that certain Norwegian and Scottish companies and production sites choose not to use lice skirts (Misund et al., 2020; A. Currie, personal communication, September 1, 2021).

Many of these issues are related to the skirt's interaction with the incoming current flow, particularly how the skirt can limit the exchange of water within the cage volume. As discussed previously, the local current flow pattern can be influenced by several factors such as hydrography, deformations of the skirt and local bathymetry. Hence, the issues can be location specific or season specific. Most of the early work has focused extensively on the impact of lice shielding skirts on dissolved oxygen (DO) levels, however there is still much more research needed on the interaction between skirts and gill health.

The concerns related to perceived increase in AGD have not been extensively researched or been scientifically verified but are mentioned briefly in the Norwegian Veterinary Institutes yearly fish health report (Hjeltnes et al., 2017; Sommerset et al., 2022). Prevalence of the marine amphizoic amoeba Paramoeba perurans has been the cause for outbreaks of AGD in several salmon producing countries (Rodger, 2014; Oldham et al., 2016). AGD triggers the development of hyperplastic lesions on the gills of the salmon, which reduces the respiratory surface area. Because the gills are vital for oxygen uptake and ion regulation, AGD increases energy requirement and decreases hypoxia tolerance in salmon (Hvas et al., 2017; Bowden et al., 2022). Furthermore, the internal cage environment has some influence on the disease. AGD infected fish have a higher mortality after being subjected to suboptimal oxygen levels (50% O2 saturation) (Fisk et al., 2002), and more clinically significant AGD is observed in heavily fouled pens or those with poorer water exchange (Rodger, 2014). Mounting of skirts around the net pen and the subsequent reduction in waterflow, could contribute to the accumulation of particles, amoebas, and algae within the cage environment, and thus increasing the prevalence of AGD and gill diseases. There are currently no studies conducted which correlates the use of skirt and higher infestation of AGD. However, sites with weak to almost no current can cause low DO levels within the cage environment (Stien et al., 2012; Remen et al., 2016), and cyclic hypoxia is known to accelerate the progress of AGD (Oldham et al., 2020). Hence ensuring proper DO levels is a necessity.

5.2. External influencing factors: Shielding skirts and dissolved oxygen

Temperature is considered the most influential controlling factor of fish metabolism since fish are ectotherms. However, oxygen availability establishes the physiological limits, meaning that activities such as locomotion, digestion, growth and reproduction, are fuelled by energy generated through aerobic metabolism, which is dependent on the availability of oxygen in the surrounding environment (Claireaux and Chabot, 2016; Neubauer and Andersen, 2019). Oxygen consumption rate (MO₂) (or aerobic metabolic rate) of salmonids is regulated by size, feeding, stress and environmental factors, where episodes of hypoxia affect the capacity for activity and locomotion, and with smaller salmon being more afflicted (Remen et al., 2016; Oldham et al., 2019). Overall, oxygen is perhaps the most critical factor in aquaculture, as adequate (upwards from approximately >80%) DO levels are vital for several performance indicators such as growth and optimization of feed conversion ratio (FCR). Low oxygen levels can affect fish growth, induce stress responses, increase vulnerability to diseases and in extreme hypoxic cases result in death (Remen et al., 2016; Burke et al., 2021).

Physical transport of oxygenated water by tidal currents and wind driven exchange is the primary factor for oxygen renewal in sea cages (Wildish et al., 1993; Johansson et al., 2006). DO levels will therefore vary both in time and space, both horizontally and vertically within sea cages. In a study by Solstorm et al. (2018) the DO levels within an unshielded cage were generally higher at deeper depths. While

horizontally, the highest DO levels were found in the upstream region of the cage, and the lowest values were recorded in the center and downstream near the cage wall (Solstorm et al., 2018). The horizontal variance was partially explained by a combination of slow ambient current speed and the presence of fish.

As the current was one of the main parameters influencing the DO levels inside unshielded cages, a physical barrier such as the lice shielding skirt is expected to influence the DO levels inside a cage as the skirt deflects the current. However, as with the shielding efficiency, although some reports and research papers note that lice shielding skirts have an adverse effect on DO levels, others see no effect over long periods (Table 4).

In a study by Stien et al. (2018) a permeable skirt was used and DO was registered daily. Initially, there was no difference in DO levels in the cages with and without skirts. However, after 2 weeks, the difference increased with average oxygen levels at 1 m depth in the cages without skirt at 104% (\pm 2%), and 83% (\pm 1%) in the cages with skirt. The levels in Stien et al. (2018) were all above 70%, hence it was concluded that the welfare of the salmon was never at risk. It should be noted that the initial biomass density at the beginning of the trial was <10 kg/m³ for all cages, and the results may therefore not be reflective of how conditions are within cages with a higher biomass.

The results were also promising in the reports by Næs et al. (2012, 2014) where no connection was made between registered average dissolved oxygen levels and skirt presence. The median levels in Næs et al. (2012) however indicate that there was some impact of the lice shielding skirt. From measurements taken at 5 m depth in the unshielded cage, the average oxygen level was 112% and minimum recorded DO was 102%, while in the shielded cages the average DO was 102% and the minimum was 67% (Næs et al., 2012).

In contrast, in Stien et al. (2012) a 3 m impermeable skirt was installed with the intention to leave it on for 6 months, but after 7 days the experiment was aborted due to low oxygen levels. Prior to the skirt being installed, the median oxygen level inside the cage was 88%. From the moment the skirt was installed the median DO level at 1.5 m began to drop, despite the reference DO outside the cage being stable at 90–100%. During the last two days of the study, median DO cycled between 45% and 90%. After removing the skirt, DO inside the cage improved and cycled between 70 and 90% for the next two days before the experiment was terminated (Stien et al., 2012).

Similar results were obtained in Jónsdóttir et al. (2021a) where a stocked sea cage had a skirt installed for two days before it had to be removed due to a steep drop in DO. A 6.7 m deep tarpaulin skirt was installed and DO was measured at 3 m depth inside and outside the cage. During the first night of the study the DO gradually dropped to a minimum of 59% in the early morning, and the skirt had to be removed. The drop in DO when the skirt was deployed occurred over a period of 6 h. A similar drop in DO was seen the next night after the skirt had been removed, however the minimum was 69% and the DO increased gradually after reaching this point. In both situations the current speed was relatively slow and appeared to be turning. It is uncertain whether the oxygen levels would have increased naturally when the skirt was deployed. However, once removed, it only took 30 min for the DO level to increase to 81% which was the same as outside the cage (Jónsdóttir et al., 2021a).

A large sudden drop in DO was also observed in Oldham et al. (2017) where the top 6 m of an experimental research scale cage of 12x12x29 m was wrapped in tarpaulin for 60 min in four trials. DO levels dropped at all depths inside the skirt volume in all four trials, compared to a 40-min period before and after installing the skirt. The largest difference in DO when the skirt was installed and removed was close to 20%.

DO levels vary horizontally inside both unshielded (Solstorm et al., 2018) and shielded cages (Stien et al., 2012). The influence of the skirt is more apparent in the vertical plane. In Stien et al. (2012) which had a 3 m impermeable skirt, DO was recorded between 1 and 20 m depth. The DO improved nearly linearly with depth from \sim 49% at 1 m depth to

Table 4

Overview over studies (both peer-reviewed and grey papers) that have studied the impact of skirts on oxygen levels. Studies with several sites may have different results at each site, so if there are cases of both a reduction and no effect, this is included in the table. A reduction does not necessarily indicate a hazardous situation for the fish, only that the oxygen inside the shielded cage was reduced compared to a reference. Comparison method for impact of shielding skirt on oxygen levels are either comparing shielded cages with unshielded cages (with/without), or with itself when the skirt is removed (on/off). The rows containing data from peer-reviewed studies are written in bold and are italicized. The other studies are grey-literature.

			Skirt and preventative measures		Trend oxygen levels			
Study	# sites	Duration	Туре	Depth	Oxygenation equipment	Reduction/ No effect	Comparison method	Min DO [%] With/W. O. Skirt
Næs et al. (2012) <i>Stien et al. (2012)</i>	1 1	30 wks (2011) 7 days (2011) 10–14 wks (2012-	Permeable Impermeable	10 m 3 m 6 & 10		Reduction <i>Reduction</i> Reduction/ No	With/ Without On/Off	67 / 102 45 / 70
Næs et al. (2014)	6	2013)	Permeable	m		effect	With/ Without	76 / 75*
Oldham et al. (2017)	1	11 days (2015)	Impermeable	6 m		Reduction	On/off	59 / 67
Stien et al. (2018)	1	15 wks (2014)	Permeable	10 m		Reduction	With/ without	~70 / 90
Jónsdóttir et al. (2021a)	1	3 days (2018)	Impermeable	6.7 m		Reduction	On/off	59 / 69

* Not same site.

 \sim 86% at 7 m depth and stayed above 80% for the remaining depth. In Klebert and Su (2020) a 10 m deep tarpaulin skirt was used, and DO was recorded at 1, 5 and 11 m depth. During the study, DO level in all sensors dropped around every midnight, but DO at 1 and 5 m were lower than at 11 m, and the minimum of 58% was recorded at 1 m depth.

This vertical variance is also seen with permeable skirts, such as in Jónsdóttir et al. (2020), where a 10 m deep permeable skirt was wrapped around a conical cage and DO record at 3, 6, 9 and 12 m depth. The study only lasted three days, but DO levels varied greatly from 98% to 52%. The lowest DO values were found at 3 and 6 m depth. However, it was only during the first two days that the sensors at 3 and 6 m depth registered lower values than at 9 and 12 m. On the last day all DO sensors inside the cage measured similar values (Jónsdóttir et al., 2020). During the study, the current conditions were relatively similar with a clear tidal pattern, but there was a pycnocline present the first day which moved upwards and eventually broke down on the last day. There was also a difference in density inside and outside the cage on the second day. The authors therefore theorized that the variation in DO levels inside the cage were caused by the interaction between density gradients, current flow and the shielding skirts (Jónsdóttir et al., 2020).

Stratified sites have reported lower mean oxygen levels inside cages, than homogenous sites in studies with unshielded cages (Johansson et al., 2007). Given that stratification can influence the current flow, it is not surprising that stratified sites can also influence the DO levels inside shielded cages. When studying two hydrographically different locations with skirts, the DO level inside the cage at the stratified location was dependent on the strength and depth of the present pycnocline (Jónsdóttir et al., 2020). The homogenous site had a much more stable DO level throughout the 4 days the study lasted. Other studies have also commented on this interaction, for instance in Stien et al. (2012) it was hypothesized by the authors that a stratification was the cause for the low DO levels inside the cage.

The low DO levels in the top 6 m in Stien et al. (2012) could also be due to the fish congregating. Based on visuals from an underwater winch camera, it was reported that the salmon tended to gather in the top 6 m of the water column. Oldham et al. (2017) used a similar tarpaulin skirt as Stien et al. (2012), however there was a clear stratification present during trials. The depth and strength of the gradient varied, with the first trial having a thermocline and a halocline near 4 m, while trial 4 had a gradient near the skirt depth of 6 m. In Oldham et al. (2017) however, the fish was observed to avoid the top 6 m, so the low DO levels inside the skirt could not be explained by the fish congregating in this region.

The combination of low current speed, turning tide and interaction with pycnoclines could explain why some studies experience low DO levels while other do not. In Klebert and Su (2020) the lowest DO levels occurred at the sensor at 1 m depth after midnight on three consecutive days. From the echosounder data the fish appeared to gather in a denser formation in the upper layers during this period. The current was turning during midnight, and current speed was generally low. It is therefore possible that the combination of high stocking density and low currents explain the dip in DO levels, similar to that observed in Jónsdóttir et al. (2021a).

Given the difference in DO levels with depth at some of these sites, it is uncertain if the salmon were ever in an unfavourable environment. It has been theorized that salmon will actively avoid low DO levels (Johansson et al., 2007), and therefore avoid the lice skirt volume. In Oldham et al. (2017) the salmon swam beneath the lice shielding skirt and were thereby avoiding low DO levels. However, no such avoidance behaviour was seen in Jónsdóttir et al. (2021a). Avoidance behaviour for low DO levels have only been observed for extreme low levels of DO (<35%) (Stehfest et al., 2017). More moderate levels are not a primary driver of avoidance behaviour, with feeding, light and temperature being more critical drivers (Oppedal et al., 2011; Burt et al., 2012; Oldham et al., 2017; Stehfest et al., 2017). For instance, in Solstorm et al. (2018), four individual fish were tagged with oxygen sensors, and all four fish experienced suboptimal DO levels, mainly during the night. It was theorized that the low DO conditions near the surface during the night were due to the fish congregating in this area, as this was the region with the lowest DO, and salmon tend to move closer to the surface after sunset (Oppedal et al., 2011).

The varying results from the studies can be attributed to some of the different parameters such as skirt type, skirt deformation, local environmental conditions, biomass density, but also size of cage and feeding activity (Alver et al., 2022). All of these factors will influence how strong a current is necessary to replenish the water within the cage. With the increased interest in offshore sites (Bjelland et al., 2015) new cage designs are emerging which are typically much larger than the conventional cage. Much of the allure with moving to these remote location is the reduced lice infestation pressure, but any plans to use shielding structures on these large cages would require careful planning and potential oxygen enhancing equipment.

5.3. Technologies for improved skirt use: Oxygenation and water circulation

To achieve optimal growth conditions when using skirts, it is necessary to ensure good DO levels inside the cage, but which strategy that achieves this will vary depending on local current conditions, topography, fish density and stratification. To improve and control the DO levels inside a cage, systems have been developed to inject pure oxygen into the water in fish cages (Bergheim et al., 2006; Srithongouthai et al., 2006). Bergheim et al. (2006) shows a significant improvement of the DO concentration by using a network of microperforated hoses for oxygen distribution. Experiments with regulation of oxygen added to the water volume significantly improved DO (Bergheim et al., 2006).

Srithongouthai et al. (2006) conducted similar experiments with a microscopic bubble generator system. The water conditions in a fish farm were monitored from June to October 2004 by a vertical profiling system operated at night. The results showed improved DO levels, but bottom water with poor oxygen content occasionally appeared in the cage volume (Srithongouthai et al., 2006).

There are no scientific publications on these systems being utilised in shielded cages, but the results were promising in non-shielded cages. In addition to these two systems, there are solutions aimed specifically at shielded cages. One of the first proposed solutions was to pump water from deeper layers below the skirt, through a tube to the surface inside the skirt, thereby bringing oxygen rich water to the top layers. The intention was to distribute the deep water on the surface and press down the oxygen poor water below the skirt (Frank and Lien, 2015).

In a full-scale situation, the deeper water might have a different density than the top layer, hence experiments in scale 1:100 were performed with various relevant density differences. The water was pumped through a tube from the bottom of the tank to the surface. The results showed no distribution effect of the deep water on the surface, independently of flow rate through the tube. Instead, the water fell to the bottom of the tank sliding along the tube. The sinking speed increasing with denser water (Frank and Lien, 2015).

Other attempts at improving the water quality within lice skirt have been made with air bubble generators. The idea was to transport pressurized air into deeper parts of the cage, below the depth of the skirt. The air is then distributed in an aerator to form bubbles. The bubbles hopefully drag the deeper water upwards in to the skirt volume, replacing the oxygen low water. There are no scientific publications that verify that this principle works, however there is a report by Aquamedic AS (Midtlyng et al., 2019) which showed no significant positive biological effect when using such bubble generators.

6. Conclusion and future work

Most of the studies aimed at documenting the shielding efficiency of lice skirts report of positive results. However, due to the large discrepancies in where, when and how the studies were executed, in addition to different methods in evaluating the lice shielding efficiency, it is difficult to draw any clear conclusion regarding the general efficiency of lice shielding skirts. The additional papers and studies which indicate no effect of lice shielding skirt, further complicates this exercise. What is clear from the studies however is that there are large local variations.

Local variations in current flow patterns can influence how the current interacts with the skirt, and thereby the lice shielding efficiency. Skirts can reduce the water exchange within the skirt by deflecting the current, forcing portion of the incoming flow to diverge around the cage. However, this effect can vary with seasonal variations at the site, for instance with change in hydrographic conditions, which can change with the influx of fresh water from nearby rivers and water run-offs. The establishment of a density gradient could prevent the upper layer of the water column from moving down and underneath the skirt, thereby reducing the water exchange further. While sites with more homogenous water columns might not struggle with insufficient water exchange, as the surface water can more easily move downwards and into the skirt, which is positive with regards to water exchange, but paradoxically might reduce the lice shielding efficiency.

Statistically documenting the general lice shielding efficiency of skirts is a difficult exercise due to the many compounding variables. Even comparing a shielded cage with an unshielded cage at the same location may be inaccurate if the current conditions are different between the two cages. Comparing two cages at one site during one season may also not be representative for the other seasons, or the next year. If the goal is to document the effect of shielding skirts at a general level, it would be optimal to study several cages with and without skirts at different sites, over all seasons. In addition, the shielded cages should be installed with a form of oxygenation so that the skirt did not have to be removed during the study, and the environment would have to be monitored simultaneously to investigate the interaction between environment and skirt.

This is a complex and resource heavy task, and given the positive results of many studies, it may be of more value for the industry if future studies and projects work towards how to optimise the usage of skirts at the individual site. To develop the optimal skirt strategy for each site, better understanding of the complex interaction between cage, skirt, environment and fish welfare must be achieved, and sufficient monitoring procedures and decision support systems must be developed. It is fundamental for each site to understand their local environmental conditions and variations within their farms to obtain positive results when using skirts. An initial step could therefore be to recommend farmers to monitor the current flow and hydrographic conditions within and outside of shielded cages, while monitoring lice numbers. This could give an indication of whether the skirt is isolating the top layer, or if the site is exposed to strong mixing of the water column, rendering the skirt useless.

Despite having been used for over a decade in Norwegian aquaculture, and having spread to other salmon farming nations, the complex interaction between skirts and potential welfare risks related to blocking the incoming current are not fully documented. Most studies have focused on low DO values, which is the most prevalent issue, however with large variations between studies and locations. A newer concern, which requires attention, is how skirts may influence gill health and increase AGD incidents. This needs further investigation to document if there is an actual link here, or rather that AGD increases due to lower DO at shielded sites and increased accumulation of particles.

CRediT authorship contribution statement

Kristbjörg Edda Jónsdóttir: Conceptualization, Data curation, Writing – original draft, Writing – review & editing, Funding acquisition, Project administration. Andreas Ugelvik Misund: Conceptualization, Data curation, Writing – original draft, Writing – review & editing. Leif Magne Sunde: Writing – review & editing. Merete Bjørgan Schrøder: Writing – review & editing. Zsolt Volent: Conceptualization, Data curation, Writing – original draft, Writing – review & editing, Visualization.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

Acknowledgements

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors. This paper was funded by SINTEF OCEAN. Special thanks to Ashleigh Currie (FiiZK) and Paul J. Midtlyng (Aquamedic AS) for taking their time to answer our questions.

Aquaculture 562 (2023) 738817

References

- Aaen, S.M., Helgesen, K.O., Bakke, M.J., Kaur, K., Horsberg, T.E., 2015. Drug resistance in sea lice: a threat to salmonid aquaculture. Trends Parasitol. 31, 72–81. https:// doi.org/10.1016/j.pt.2014.12.006.
- Abolofia, J., Asche, F., Wilen, J.E., 2017. The cost of lice: quantifying the impact of Salmon lice on farmed Salmon. Mar. Resour. Econ. 32.
- Allen, S.E., Lewis, A.G., 2013. Lepeophtheirus salmonis (Krøyer, 1837): second nauplius and copepodid locomotor appendages, surface areas and possible appendage functions. Crustaceana 86 (13–14), 1695–1710.
- Alver, M.O., Føre, M., Alfredsen, J.A., 2022. Predicting oxygen levels in Atlantic salmon (Salmo salar) sea cages. Aquaculture 548, 737720.
- Barrett, L.T., Overton, K., Stien, L.H., Oppedal, F., Dempster, T., 2020a. Effect of cleaner fish on sea lice in Norwegian salmon aquaculture: a national scale data analysis. Int. J. Parasitol. 50, 787–796.
- Barrett, L.T., Oppedal, F., Robinson, N., Dempster, T., 2020b. Prevention not cure: a review of methods to avoid sea lice infestations in salmon aquaculture. Rev. Aquac. 12, 2527–2543. https://doi.org/10.1111/raq.12456.
- Bechmann, R.K., Arnberg, M., Gomiero, A., Westerlund, S., Lyng, E., Berry, M., Burridge, L.E., 2019. Gill damage and delayed mortality of northern shrimp (Pandalus borealis) after short time exposure to anti-parasitic veterinary medicine containing hydrogen peroxide. Ecotoxicol. Environ. Saf. 180, 473–482. https://doi. org/10.1016/j.ecoenv.2019.05.045.
- Bechmann, R.K., Arnberg, M., Bamber, S., Lyng, E., Westerlund, S., Rundberget, J.T., Burridge, L., 2020. Effects of exposing shrimp larvae (Pandalus borealis) to aquaculture pesticides at field relevant concentrations, with and without food limitation. Aquat. Toxicol. 222, 105453 https://doi.org/10.1016/j. aquatox.2020.105453.
- Bergheim, A., Gausen, M., Næss, A., Hølland, P.M., Krogedal, P., Crampton, V., 2006. A newly developed oxygen injection system for cage farms. Aquac. Eng. 34, 40–46. https://doi.org/10.1016/j.aquaeng.2005.04.003.
- Bi, C.W., Zhao, Y.P., Dong, G.H., Xu, T.J., Gui, F.K., 2013. Experimental investigation of the reduction in flow velocity downstream from a fishing net. Aquac. Eng. 57, 71–81.
- Bjelland, H.V., Føre, M., Lader, P., Kristiansen, D., Holmen, I.M., Fredheim, A., et al., 2015. Exposed aquaculture in Norway. In: OCEANS 2015-MTS/IEEE Washington. IEEE, pp. 1–10.
- Bowden, A.J., Adams, M.B., Andrewartha, S.J., Elliott, N.G., Frappel, P.B., Clark, T.D., 2022. Amoebic Gill Disease Increases Energy Requirements and Decreases Hypoxia Tolerance in Atlantic Salmon (*Salmo salar*) Smolts. Comparative Biochemistry and physiology Part A: Molecular & Integrative Physiology, p. 265.
- Brooker, A.J., Skern-Mauritzen, R., Bron, J.E., 2018. Production, mortality, and infectivity of planktonic larval sea lice, Lepeophtheirus salmonis (Krøyer, 1837): current knowledge and implications for epidemiological modelling. ICES J. Mar. Sci. 75 (4), 1214–1234.
- Bui, S., Stien, L.H., Nilsson, J., Trengereid, H., Oppedal, F., 2020. Efficiency and welfare impact of long-term simultaneous in situ management strategies for salmon louse reduction in commercial sea cages. Aquaculture 520, 734934.
- Burke, M., Grant, J., Filgueira, R., Stone, T., 2021. Oceanographic processes control dissolved oxygen variability at a commercial Atlantic salmon farm: application of a real-time sensor network. Aquaculture 533, 736143.
- Burt, K., Hamoutene, D., Mabrouk, G., Lang, C., Puestow, T., Drover, D., Page, F., 2012. Environmental conditions and occurrence of hypoxia within production cages of Atlantic salmon on the south coast of Newfoundland. Aquac. Res. 43 (4), 607–620.
- Claireaux, G., Chabot, D., 2016. Responses by fishes to environmental hypoxia: integration through Fry's concept of aerobic metabolic scope. J. Fish Biol. 88 (1), 232–251.
- Coates, A., Phillips, B.L., Oppedal, F., Bui, S., Overton, K., Dempster, T., 2020. Parasites under pressure: salmon lice have the capacity to adapt to depth-based preventions in aquaculture. Int. J. Parasitol. 50 (10–11), 865–872.
- Coates, A., Phillips, B.L., Bui, S., Oppedal, F., Robinson, N.A., Dempster, T., 2021. Evolution of salmon lice in response to management strategies: a review. Rev. Aquac. 13 (3), 1397–1422.
- Costello, M.J., 2006. Ecology of sea lice parasitic on farmed and wild fish. Trends Parasitol. 22 (10), 475–483.
- Crosbie, T., Wright, D.W., Oppedal, F., Johnsen, I.A., Samsing, F., Dempster, T., 2019. Effects of step salinity gradients on salmon lice larvae behaviour and dispersal. Aquacul. Environ. Interact. 11, 181–190.
- Crosbie, T., Wright, D.W., Oppedal, F., Dalvin, S., Myksvoll, M.S., Dempster, T., 2020. Impact of thermoclines on the vertical distribution of salmon lice larvae. Aquacul. Environ. Interact. 12, 1–10.
- DeCew, J., Fredriksson, D., Lader, P., Chambers, M., Howell, W., Osienki, M., Høy, E., 2013. Field measurements of cage deformation using acoustic sensors. Aquac. Eng. 57, 114–125. https://doi.org/10.1016/j.aquaeng.2013.09.006.
- Dempster, T., Korsøen, Ø., Folkedal, O., Juell, J.E., Oppedal, F., 2009. Submergence of Atlantic salmon (Salmo salar L.) in commercial scale sea-cages: a potential shortterm solution to poor surface conditions. Aquaculture 288 (3–4), 254–263.
- Finstad, B., Bjørn, P.A., Grimnes, A., Hvidsten, N.A., 2000. Laboratory and field investigations of salmon lice [Lepeophtheirus salmonis (Krøyer)] infestation on Atlantic salmon (Salmo salar L.) post-smolts. Aquac. Res. 31 (11), 795–803.
- Finstad, B., Bjørn, P.A., Todd, C.D., Whoriskey, F., Gargan, P.G., Forde, G., Revie, C.W., 2011. The effect of sea lice on Atlantic salmon and other salmonid species. In: Atlantic Salmon Ecology. Wiley-Blackwell, pp. 253–276. https://doi.org/10.1002/ 9781444327755.ch10.
- Fisk, D.M., Powell, M.D., Nowak, B.F., 2002. The effect of amoebic gill disease and hypoxia on survival and metabolic rate of Atlantic salmon (Salmo salar). Bull. Eur. Assoc. Fish Pathol. 22 (3), 190–194.

- Forseth, T., Barlaup, B.T., Finstad, B., Fiske, P., Gjøsæter, H., Falkegård, M., Wennevik, V., 2017. The major threats to Atlantic salmon in Norway. ICES J. Mar. Sci. 74 (6), 1496–1513.
- Frank, K., Lien, A.M., 2015. Permaskjørt og merdmiljø. SINTEF Report No. A26686 (ISBN: 978-82-14-05867-3).
- Frank, K., Gansel, L.G., Lien, A.M., Birkevold, J., 2015. Effect of a shielding skirt for prevention of sea lice on the flow past stocked Salmon fish cages. J. Offshore Mech. Arct. Eng. 137 (1), 011201, 6 pages. https://doi.org/10.1115/1.4028260.
- Gansel, L.C., Rackebrandt, S., Oppedal, F., McClimans, T.A., 2014. Flow fields inside stocked fish cages and the near environment. J. Offshore Mech. Arctic Eng. 136, 031201. https://doi.org/10.1115/1.4027746.
- Garcia de Leaniz, C., Gutierrez Rabadan, C., Barrento, S.I., Stringwell, R., Howes, P.N., Whittaker, B.A., et al., 2022. Addressing the welfare needs of farmed lumpfish: knowledge gaps, challenges and solutions. Rev. Aquac. 14 (1), 139–155.
- Geitung, L., Oppedal, F., Stien, L.H., Dempster, T., Karlsbakk, E., Nola, V., Wright, D.W., 2019. Snorkel Sea-cage technology decreases salmon louse infestation by 75% in a full-cycle commercial test. Int. J. Parasitol. 49 (11), 843–846.
- Geitung, L., Wright, D.W., Oppedal, F., Stien, L.H., Vågseth, T., Madaro, A., 2020. Cleaner fish growth, welfare and survival in Atlantic salmon sea cages during an autumn-winter production. Aquaculture 528, 735623.
- Gentry, K., Bui, S., Oppedal, F., Dempster, T., 2020. Sea lice prevention strategies affect cleaner fish delousing efficacy in commercial Atlantic salmon sea cages. Aquacul. Environ. Interact. 12, 67–80.
- Grøntvedt, R., Kristoffersen, A., 2015. Permaskjørt kan redusere påslag av lakselus. Delrapport Permaskjørt-prosjektet A5. Norwegian Vet. Instit. Rep. Ser. 2 (2015), 1–28.
- Grøntvedt, R.N., Kristoffersen, A.B., Jansen, P.A., 2018. Reduced exposure of farmed salmon to salmon louse (Lepeophtheirus salmonis L.) infestation by use of plankton nets: estimating the shielding effect. Aquaculture 495, 865–872.
- Hamre, L.A., Eichner, C., Caipang, C.M.A., Dalvin, S.T., Bron, J.E., Nilsen, F., et al., 2013. The salmon louse Lepeophtheirus salmonis (Copepoda: Caligidae) life cycle has only two chalimus stages. PLoS One 8 (9), e73539.
- Heuch, P.A., Parsons, A., Boxaspen, K., 1995. Diel vertical migration: a possible hostfinding mechanism in salmon louse (Lepeophtheirus salmonis) copepodids? Can. J. Fish. Aquat. Sci. 52, 681–689. https://doi.org/10.1139/f95-069.
- Hevrøy, E., Boxaspen, K., Oppedal, F., Taranger, G., Holm, J.C., 2003. The effect of artificial light treatment and depth on the infestation of the sea louse lepeophtheirus salmonis on Atlantic salmon (Salmo salar L.) culture. Aquaculture 220 (1–4), 1–14. https://doi.org/10.1016/s0044-8486(02)00189-8.
- Hjeltnes, B., Bornø, G., Jansen, M.D., Haukaas, A., Walde, C., 2017. The Health Situation in Norwegian Aquaculture 2016. Norwegian Veterinary Institute, Oslo.
 Hvas, M., Karlsbakk, E., Mæhle, S., Wright, S.W., Oppedal, F., 2017. The gill parasite
- Hvas, M., Karlsbakk, E., Mæhle, S., Wright, S.W., Oppedal, F., 2017. The gill parasite Paramoeba peruans compromises aerobic scope, swimming capacity and ion balance in Atlantic salmon. Conserv. Physiol. 5 (1).
- Imberger, J., 2012. Environmental Fluid Dynamics: Fluid Processes, Flow Scales and Processes, and Equations of Motion. Elsevier/Academic Press.
- Imsland, A.K., Reynolds, P., Eliassen, G., Hangstad, T.A., Foss, A., Vikingstad, E., Elvegård, T.A., 2014. The use of lumpfish (Cyclopterus lumpus l.) to control sea lice (lepeophtheirus salmonis krøyer) infestations in intensively farmed Atlantic salmon (salmo Salar l.). Aquaculture 424, 18–23. https://doi.org/10.1016/j. aquaculture.2013.12.033.
- Imsland, A.K.D., Hanssen, A., Nytrø, A.V., Reynolds, P., Jonassen, T.M., Hangstad, T.A., Mikalsen, B., 2018. It works! Lumpfish can significantly lower sea lice infestation in large-scale salmon farming. Biology Open 7 (9). https://doi.org/10.1242/ bio.036301.
- Iversen, A., Hermansen, Ø., Nystøyl, R., Rolland, K.H., Garshol, L.D., 2019. Konkurranseevne for norsk oppdrettslaks: Kostnader og kostnadsdrivere i konkurrentland (In English: Competitiveness of Norwegian farmed salmon: Costs and Cost Drivers In Competitor Countries). Rapport 28/2019, Nofima.
- Jenssen, E., 2016. Er dette et skip, en rigg eller et oppdrettsanlegg?(in Norwegian). Teknisk Ukeblad. https://www.tu.no/artikler/er-dette-et-skip-en-rigg-eller-et-opp drettsanlegg/350710. (Accessed 12 October 2021).
- Johansson, D., Ruohonen, K., Kiessling, A., Oppedal, F., Stiansen, J.E., Kelly, M., Juell, J. E., 2006. Effect of environmental factors on swimming depth preferences of Atlantic salmon (salmo Salar l.) and temporal and spatial variations in oxygen levels in sea cages at a fjord site. Aquaculture 254 (1–4), 594–605. https://doi.org/10.1016/j. aquaculture.2005.10.029.
- Johansson, D., Juell, J.E., Oppedal, F., Stiansen, J.E., Ruohonen, K., 2007. The influence of the pycnocline and cage resistance on current flow, oxygen flux and swimming behaviour of Atlantic salmon (Salmo salar L.) in production cages. Aquaculture 265 (1–4), 271–287.
- Johansson, D., Laursen, F., Fernö, A., Fosseidengen, J.E., Klebert, P., Stien, L.H., Oppedal, F., 2014. The interaction between water currents and salmon swimming behaviour in sea cages. PLoS One 9 (5), e97635. https://doi.org/10.1371/journal. pone.0097635.
- Jónsdóttir, K.E., Volent, Z., Alfredsen, J.A., 2020. Dynamics of dissolved oxygen inside salmon seacages with lice shielding skirts at two hydrographically different sites. Aquacul. Environ. Interact. 12, 559–570.
- Jónsdóttir, K.E., Volent, Z., Alfredsen, J.A., 2021a. Current flow and dissolved oxygen in a full-scale stocked fish-cage with and without lice shielding skirts. Appl. Ocean Res. 108, 102509.
- Jónsdóttir, K.E., Klebert, P., Volent, Z., Alfredsen, J.A., 2021b. Characteristic current flow through a stocked conical sea-cage with permeable lice shielding skirt. Ocean Eng. 223, 108639.

Klebert, P., Su, B., 2020. Turbulence and flow field alterations inside a fish sea cage and its wake. Appl. Ocean Res. 98, 102113 https://doi.org/10.1016/j. apor.2020.102113.

- Klebert, P., Lader, P., Gansel, L., Oppedal, F., 2013. Hydrodynamic interactions on net panel and aquaculture fish cages: A review. Ocean Eng. 58, 260–274. https://doi. org/10.1016/j.oceaneng.2012.11.006.
- Klebert, P., Patursson, Ø., Endresen, P.C., Rundtop, P., Birkevold, J., Rasmussen, H.W., 2015. Three-dimensional deformation of a large circular flexible sea cage in high currents: field experiment and modeling. Ocean Eng. 104, 511–520.
- Lien, A.M., Høy, E., 2011. Skjørt for skjerming mot lus i laksemerd. SINTEF Fiskeri og Havbruk AS, Report.
- Lien, A.L., Volent, Z., 2012. Deformasjon av not og Permaskjørt og krefter på fortøyning. SINTEF report no. A23184 (ISBN: 978-82-14-05445-3).
- Lien, A.L., Volent, Z., Jensen, Ø., Lader, P., Sunde, L.M., 2014. Shielding skirt for prevention of salmon lice (*Lepeophtheirus salmonis*) infestation on Atlantic salmon (*Salmo salar L.*) in cages – A scaled model experimental study on net and skirt deformation, total mooring load, and currents. Aquac. Eng. 58, 1–10. https://doi. org/10.1016/j.aquaeng.2013.11.003.
- Løland, G., 1993. Current forces on, and water flow through and around, floating fish farms. Aquac. Int. 1 (1), 72–89. https://doi.org/10.1007/bf00692665.
- Midtlyng, P.J., van Velzen, T., Aunsmo, A., Førde, H., Stormoen, M., Storsul, Y., Munkeby, M., 2019. Faglig sluttrapport for FHF-prosjektet 901543: Dokumentasjon av lusebeskyttelse med Midt-Norsk Ringen. Aquamedic Rep. 1, 1–25. https://www. fhf.no/prosjekter/prosjektbasen/901453/.
- Midt-Norsk Havbruk, A.S., 2020. Deep Lice Skirts Few Salmon Lice. https://aquatraz. com/en/launching-aquatraz-g3/.
- Misund, A.U., 2019. From a natural occurring parasitic organism to a management object: historical perceptions and discourses related to salmon lice in Norway. Mar. Policy 99, 400–406.
- Misund, A., Volent, Z., Jónsdóttir, K.E., Sunde, L.M., 2020. Hvordan forholder oppdrettere seg til skjermingsteknologi mot lakselus?. In: Norsk Fiskeoppdrett nr. 9, side 52-57, 2020.
- Næs, M., Heuch, P.A., Mathisen, R., 2012. Bruk av "luseskjørt" for å redusere påslag av lakselus Lepeophtheirus salmonis (Krøyer) på oppdrettslaks. Faglig rapport.
- Næs, M., Grøntvedt, R., Kristoffersen, A., Johansen, B., 2014. Feltutprøving av planktonduk som skjerming rundt oppdrettsmerder for å redusere påslag av lakselus (lepeophtheirus salmonis). Faglig Rapp. 4.
- Neubauer, P., Andersen, K.H., 2019. Thermal performance of fish is explained by an interplay between physiology, behaviour and ecology. Conserv. Physiol. 7 (1), coz025. https://doi.org/10.1093/conphys/coz025.
- Noble, C., Gismervik, K., Iversen, M.H., Kolarevic, J., Nilsson, J., Stien, L.H., Turnbull, J. F. (Eds.), 2018. Welfare Indicators for Farmed Atlantic Salmon: Tools for Assessing Fish Welfare. Report. NOFIMA.
- Norwegian Ministry of Trade, Industry and Fisheries, 2012. Forskrift om bekjempelse av lakselus i akvakulturanlegg. Norwegian Ministry of Trade and Fisheries (in Norwegian).
- Oldham, T., Rodger, H., Nowak, B.F., 2016. Incidence and distribution of amoebic gill disease (AGD)—an epidemiological review. Aquaculture 457, 35–42.
- Oldham, T., Dempster, T., Fosse, J.O., Oppedal, F., 2017. Oxygen gradients affect behaviour of caged Atlantic salmon Salmo salar. Aquacul. Environ. Interact. 9, 145–153.
- Oldham, T., Nowak, B., Hvas, M., Oppedal, F., 2019. Metabolic and functional impacts of hypoxia vary with size in Atlantic salmon. Comp. Biochem. Physiol. A Mol. Integr. Physiol. 231, 30–38.
- Oldham, T., Dempster, T., Crosbie, P., Adams, M., Nowak, B., 2020. Cyclic hypoxia exposure accelerates the progression of amoebic gill disease. Pathogens 9 (8), 597.
- Oppedal, F., Dempster, T., Stien, L.H., 2011. Environmental drivers of Atlantic salmon behaviour in sea-cages: a review. Aquaculture 311 (1–4), 1–18.
 Oppedal, F., Samsing, F., Dempster, T., Wright, D.W., Bui, S., Stien, L.H., 2017. Sea lice
- infestation levels decrease with deeper 'snorkel'barriers in Atlantic salmon seacages. Pest Manag. Sci. 73 (9), 1935–1943.
- Overton, K., Dempster, T., Oppedal, F., Kristiansen, T.S., Gismervik, K., Stien, L.H., 2019. Salmon lice treatments and salmon mortality in Norwegian aquaculture: a review. Rev. Aquac. 11 (4), 1398–1417. https://doi.org/10.1111/raq.12299.
- Overton, K., Barrett, L.T., Oppedal, F., Kristiansen, T.S., Dempster, T., 2020. Sea lice removal by cleaner fish in salmon aquaculture: a review of the evidence base. Aquacul. Environ. Interact. 12, 31–44. https://doi.org/10.3354/aei00345.

- Patursson, O., 2008. Flow through and around Fish Farming Nets. Thesis. University of New Hampshire.
- Remen, M., Sievers, M., Torgersen, T., Oppedal, F., 2016. The oxygen threshold for maximal feed intake of Atlantic salmon post-smolts is highly temperaturedependent. Aquaculture 464, 582–592.
- Rodger, H.D., 2014. Amoebic gill disease (AGD) in farmed salmon (Salmo salar) in Europe. Fish Vet. J. 14, 16–27.
- Solstorm, D., Oldham, T., Solstorm, F., Klebert, P., Stien, L.H., Vågseth, T., Oppedal, F., 2018. Dissolved oxygen variability in a commercial sea-cage exposes farmed Atlantic salmon to growth limiting conditions. Aquaculture 486, 122–129.
- Sommerset, I., Walde, C.S., Bang Jensen, B., Wiik-Nielsen, J., Bornø, G., Oliveira, V.H.S., Brun, E., 2022. *The Health Situation in Norwegian Aquaculture* (Fiskehelserapporten) 2021. Norwegian Veterinary Institute, Oslo.
- Srithongouthai, S., Endo, A., Inoue, A., Kinoshita, K., Yoshioka, M., Sato, A., Tsutsumi, H., 2006. Control of dissolved oxygen levels of water in net pens for fish farming by a microscopic bubble generating system. Fish. Sci. 72 (3), 485–493. https://doi.org/10.1111/j.1444-2906.2006.01176.x.
- Stehfest, K.M., Carter, C.G., McAllister, J.D., Ross, J.D., Semmens, J.M., 2017. Response of Atlantic salmon Salmo salar to temperature and dissolved oxygen extremes established using animal-borne environmental sensors. Sci. Rep. 7 (1), 1–10.
- Stien, L.H., Nilsson, J., Hevrøy, E.M., Oppedal, F., Kristiansen, T.S., Lien, A.M., Folkedal, O., 2012. Skirt around a salmon sea cage to reduce infestation of salmon lice resulted in low oxygen levels. Aquac. Eng. 51, 21–25.
- Stien, L.H., Dempster, T., Bui, S., Glaropoulos, A., Fosseidengen, J.E., Wright, D.W., Oppedal, F., 2016. Snorkel'sea lice barrier technology reduces sea lice loads on harvest-sized Atlantic salmon with minimal welfare impacts. Aquaculture 458, 29–37.
- Stien, L.H., Lind, M.B., Oppedal, F., Wright, D.W., Seternes, T., 2018. Skirts on salmon production cages reduced salmon lice infestations without affecting fish welfare. Aquaculture 490, 281–287.
- Svåsand, T., Grefserud, E.S., Karlsen, Ø., Kvamme, B.O., Glover, K., Husa, V., Kristiansen, T.S. (Eds.), 2017. Risikorapport norsk fiskeoppdrett 2017. Report. Havforskningsinstituttet.
- Szetey, A., Wright, D.W., Oppedal, F., Dempster, T., 2021. Salmon lice nauplii and copepodids display different vertical migration patterns in response to light. Aquacul. Environ. Interact. 13, 121–131.
- Thorstad, E.B., Todd, C.D., Uglem, I., Bjørn, P.A., Gargan, P.G., Vollset, K.W., Finstad, B., 2015. Effects of salmon lice Lepeophtheirus salmonis on wild sea trout Salmo trutta a literature review. Aquacul. Environ. Interact. 7 (2), 91–113.
- Thorvaldsen, T., Holmen, I.M., Moe, H.K., 2015. The escape of fish from Norwegian fish farms: causes, risks and the influence of organisational aspects. Mar. Policy 55, 33–38.
- Volent, Z., Bekkevold, A., 2017. Erfaringsdelingsseminar med demoforsøk. SINTEF report no. OC2017 A-155 (ISBN: 978–82–7174-309-9).
- Volent, Z., Jónsdóttir, K.E., 2019. Strategi lakselus 2017: Luseskjørt som ikkemedikamentell metode for forebygging og kontroll av lakselus–Utvikling av kunnskap om miljøforhold for økt effekt og redusert risiko (SKJERMTEK) FHF prosjekt nr.: 901396.
- Volent, Z., Birkevold, J., Stahl, A., Lien, A.M., Sunde, L.M., Lader, P., 2017. Experimental study of installation procedure and volume estimation of tarpaulin for chemical treatment of fish in floating cages. Aquac. Eng. 78, 114–129. https://doi.org/ 10.1016/j.aquaeng.2017.05.003.
- Whyte, S.K., Jimenez, D., Revie, C.W., Hammel, K.L., 2016. Multivariate evaluation of the effectiveness of delousing treatment efficacy of azamethiphos (Salmosan®) against the salmon louse (Lepeophtheirus salmonis) in Atlantic salmon (Salmo salar) using wellboat, skirt and tarpaulin treatment modalities. Aquaculture 450, 301–307.
- Wildish, D., Keizer, P., Wilson, A., Martin, J., 1993. Seasonal changes of dissolved oxygen and plant nutrients in seawater near salmonid net pens in the macrotidal bay of fundy. Can. J. Fish. Aquat. Sci. 50, 303–311. https://doi.org/10.1139/f93-035.
- fundy. Can. J. Fish. Aquat. Sci. 50, 303–311. https://doi.org/10.1139/f93-035. Winthereig-Rasmussen, H., Simonsen, K., Patursson, Ø., 2016. Flow through fish farming sea cages: comparing computational fluid dynamics simulations with scaled and fullscale experimental data. Ocean Eng. 124, 21–31. https://doi.org/10.1016/j. oceaneng.2016.07.027.
- Wright, D.W., Stien, L.H., Dempster, T., Vågseth, T., Nola, V., Fosseidengen, J.E., Oppedal, F., 2017. Snorkel'lice barrier technology reduced two co-occurring parasites, the salmon louse (Lepeophtheirus salmonis) and the amoebic gill disease causing agent (Neoparamoeba perurans), in commercial salmon sea-cages. Preventive Veterinary Medicine 140, 97–105.